Author’s response to reviews

Title: A putative role for the aryl hydrocarbon receptor (AHR) gene in a patient with cyclical Cushing’s disease

Authors:

Sunita De Sousa (Sunita.DeSousa@sa.gov.au)
Jim Manavis (jim.manavis@adelaide.edu.au)
Jinghua Feng (jinghua.feng@adelaide.edu.au)
Paul Wang (paul.wang@adelaide.edu.au)
Andreas Schreiber (andreas.schreiber@adelaide.edu.au)
Hamish Scott (Hamish.Scott@sa.gov.au)
David Torpy (David.Torpy@sa.gov.au)

Version: 4 Date: 18 Jan 2020

Author’s response to reviews:

Dear staff,

The following two points have been addressed:

1. In the "Consent for publication" section, please clarify if written informed consent for publication was obtained from all individuals.

2. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

We hope that the separate figure files are of satisfactory quality. If not, please advise the specific features that are problematic and we will endeavour to fix these asap.

Many thanks again,

Sunita De Sousa
(on behalf of all authors)